Author's response to reviews

Title: Developing a decision aid to guide public sector health policy decisions: A study protocol

Authors:

Peggy Tso (Peggy.tso@utoronto.ca)
Anthony J Culyer (tony.culyer@utoronto.ca)
Melissa Brouwers (mbrouwer@mcmaster.ca)
Mark J Dobrow (mark.dobrow@utoronto.ca)

Version: 5 Date: 23 March 2011

Author's response to reviews: see over
March 22, 2011

Paul M Wilson
Associate Editor
Implementation Science

Dear Mr Paul M Wilson,

We have submitted a revised draft of our manuscript entitled ‘Developing a decision aid to guide public sector health policy decisions: A study protocol’. The following revision was made in response to the reviewer’s comments:

1) Minor essential revisions: The manuscript has one small edit needed on line 125 where the word “to” has been deleted – I think it was meant to be left in.

   The word “to” has been replaced in the manuscript to improve the grammar of the sentence.

We trust that we have satisfactorily addressed the issues raised by the reviewer and thank them for their assistance in improving this paper. On behalf of my co-authors, we thank you for your acceptance for publication.

Yours sincerely,

Peggy Tso, BHSc, MSc Candidate
Department of Health Policy Management & Evaluation
University of Toronto

Anthony J Culyer, CBE, BA, Hon DEcon, Hon FRCP, FRSA, FMedSci
Professor
Department of Health Policy, Management and Evaluation
University of Toronto

Melissa Brouwers, BSc, MA, PhD
Associate Professor
Department of Oncology
McMaster University

Mark Dobrow, MSc, PhD
Assistant Professor
Department of Health Policy, Management and Evaluation
University of Toronto